Author’s response to reviews

Title: Reconsidering low dose aspirin therapy for cardiovascular disease: an integrative approach study protocol

Authors:

Brittany Folks (Brittany.Folks@ucdenver.edu)
William G. LeBlanc (William.LeBlanc@ucdenver.edu)
Elizabeth W. Staton (Elizabeth.Staton@ucdenver.edu)
Wilson D. Pace (Wilson.Pace@ucdenver.edu)

Version: 3 Date: 5 January 2011

Author’s response to reviews:

Dear Editors:

Thank you for the opportunity to revise our manuscript. Based on the response from Dr. Foy (dated November 29, 2010), we have provided documentation that the study protocol was approved by our institutional review board/humas subjects board. The project is unfunded, so no funding documentation is available. Further, we have revised the manuscript in accordance with the CONSORT statement, including providing a CONSORT diagram, information on power calculations, and analysis plans to address clustering. We have also included the CONSORT checklist.

We look forward to your decision regarding this manuscript.

Sincerely,

Brittany Folks, BS
William LeBlanc, PhD
Elizabeth W. Staton, MSTC
Wilson D. Pace, MD